

LJMU Research Online

Tempest, N, France-Ratcliffe, M, Al-Lamee, H, Oliver, ER, Slaine, EE, Drakeley, AJ, Sprung, VS and Hapangama, DK

Habitual physical activity levels in women attending the one stop infertility clinic: A prospective cross-sectional observational study.

https://researchonline.ljmu.ac.uk/id/eprint/18932/

Article

Citation (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Tempest, N, France-Ratcliffe, M, Al-Lamee, H, Oliver, ER, Slaine, EE, Drakeley, AJ, Sprung, VS ORCID logoORCID: https://orcid.org/0000-0002-2666-4986 and Hapangama, DK (2022) Habitual physical activity levels in women attending the one stop infertility clinic: A prospective cross-

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact researchonline@ljmu.ac.uk

http://researchonline.ljmu.ac.uk/

RESEARCH

Habitual physical activity levels in women attending the one-stop infertility clinic: a prospective cross-sectional observational study

Nicola Tempest¹,^{2,3}, Madeleine France-Ratcliffe⁴, Hannan Al-Lamee^{1,3}, Evie R Oliver¹, Emily E Slaine¹, Andrew J Drakeley^{2,3}, Victoria S Sprung⁴ and Dharani K Hapangama¹,²

¹Centre for Women's Health Research, Department of Women's and Children's Health, Institute of Life Course and Medical Sciences, University of Liverpool, Member of Liverpool Health Partners, Liverpool, UK

²Liverpool Women's NHS Foundation Trust, Member of Liverpool Health Partners, Liverpool, UK ³Hewitt Centre for Reproductive Medicine, Liverpool Women's NHS Foundation Trust, Liverpool, UK ⁴Research Institute for Sport & Exercise Sciences, Liverpool John Moores University, Liverpool, UK

Correspondence should be addressed to N Tempest: ntempest@liverpool.ac.uk

Abstract

Optimisation of lifestyle factors such as smoking and alcohol is encouraged to improve fecundability rates in the fertility setting. Currently, routine fertility consultations do not involve counselling or imparting advice regarding habitual physical activity (PA) and/or structured exercise, despite data showing that vigorous PA can be associated with delayed time to pregnancy. Therefore, this study aimed to determine habitual PA in a sample of women attending the one-stop infertility (OSI) clinic. In total, 250 women attending a large tertiary level NHS fertility unit prospectively completed a questionnaire anonymously over a period of 9 months. The participants (mean age: 34 ± 5 years, mean BMI: 29 ± 7 kg/m²) habitual PA levels varied from vigorous exercise on ≥ 5 days/week (8%, n = 17) to no moderate or high-intensity activities across the whole week (66%, n = 29). A majority of women reported no structured exercise (72%, n = 179). No association was identified between any domain of PA and BMI, age, alcohol units, regular periods, or time spent trying to conceive (P > 0.05). The participants habitual PA levels varied widely and no association between any domain of PA and background of the women was identified. No existing evidence and/or guidelines to explicitly inform women attempting to conceive regarding recommended PA levels are available, despite PA being a modifiable, affordable, and feasible lifestyle choice with the possible potential to improve fertility. A large-scale, clinical trial assessing the effects of PA on fecundability is warranted to gain insight into the potential of this lifestyle factor to improve fertility outcomes and to explore the underlying biological mechanisms involved.

Lay summary

Infertility affects approximately one in seven couples with many and varied causes, including lifestyle factors such as smoking, alcohol, and diet. Lifestyle changes are low-cost unimposing options to implement in routine fertility care. Information on regular physical activity is not currently enquired from women and no agreement regarding the best levels of exercise exists for fertility patients. In this study, we aimed to determine the exercise habits of women attending the OSI clinic. In total. 250 women attending OSI clinic over a period of 9 months completed a questionnaire collecting data on their exercise habits. The levels of physical activity performed varied widely from vigorous exercise on \geq 5 days/week to no moderate or high-intensity activities across the whole week. A majority of women did no structured exercise (72%). These novel data highlight the variations in form, type and intensity of exercise women who attend OSI clinics perform.



This work is licensed under a Creative Commons Attribution 4.0 International License. **Keywords:** ► physical activity ► exercise ► fertility ► infertility ► conception

currently acknowledged in the fertility setting and has the potential to improve fertility.

Reproduction and Fertility (2022) 3 231-237

Introduction

Infertility is defined as a reproductive disease, described as the failure to achieve a clinical pregnancy within 12 months or more, of regular unprotected sexual intercourse (Zegers-Hochschild et al. 2009). Approximately one in seven couples are likely to experience difficulty conceiving (NICE 2017, NHS). The main causes often overlap and, in the UK, approximately 40% of cases identify disorders in both men and women (NICE 2017). They include unexplained infertility (25%), ovulatory disorders (25%), tubal damage (20%), factors in the male causing infertility (30%), and uterine or peritoneal disorders (10%) (NICE 2017). Infertility causes a huge burden on those couples affected by both psychological and physical strain (Schmidt 2006). Some lifestyle and recreational factors, such as alcohol/drug consumption, smoking, and hobbies, can affect fecundability rates and couples are counselled regarding the optimisation of these, accordingly (Britton 2016, NICE 2017, Ricci et al. 2017, Ayton 2019, Imterat et al. 2019, Sermondade et al. 2019).

Currently, routine fertility consultations do not involve counselling or imparting advice regarding habitual physical activity (PA) and/or structured exercise despite data from two large observational fecundability studies showing that vigorous PA was associated with delayed time to pregnancy in women with a normal BMI not currently seeking fertility care (Wise et al. 2012, McKinnon et al. 2016). Numerous other studies have examined the association between exercise and specific fertility outcomes, but the available evidence remains equivocal (Bullen et al. 1985, Ahrens et al. 2014, Mena et al. 2020). This variability within the literature holds true for women living with polycystic ovarian syndrome (PCOS), a heterogenous condition frequently associated with ovulatory disorders. It has been observed that increased PA and exercise in these women improve overall health outcomes and symptom presentation, but findings related to fertility outcomes are contradictory and relate more to weight status than PA levels (Harrison et al. 2011, Sprung et al. 2013, Kite et al. 2019, Stepto et al. 2019, Dos Santos et al. 2020). With this in mind, a change in habitual PA is potentially a cost-effective, minimally invasive adjunct, associated with general health benefits, for women to consider as part of their routine fertility care.

232

Therefore, the aim of the current study was to determine habitual PA in a sample of women attending the one-stop infertility (OSI) clinic. Furthermore, we sought to assess the willingness of these women to participate in further research examining the influence of PA on fertility outcomes.

Materials and methods

The questionnaire was distributed to 350 non-pregnant females, aged 34 ± 5 years, who attended the OSI clinic at The Hewitt Fertility Centre, Liverpool (a specialist tertiary referral fertility centre), between November 2020 and July 2021. Eligibility was based on three simple criteria; (i) non-pregnant women; (ii) aged 18–42 years; and (iii) attending the clinic for fertility consultation.

Ethical approval

This observational study conformed to the Declaration of Helsinki and was approved by the London-Bloomsbury Research Ethics Committee (REC: 20/PR/0458). All participants were informed of the protocol verbally and in writing prior to completion of the questionnaire. This detailed that consent is assumed on the voluntary return of the completed questionnaire (which did not contain any personal identifiable data).

Demographics

Women who completed the International Physical Activity Questionnaire Short Form (IPAQ-SF) self-reported their age, height, weight, ethnicity, smoking status, alcohol



consumption, occurrence of regular periods, gravidity, parity, length of time trying to conceive, diagnosis of fertility pathology, previous fertility treatment, medical conditions, and regular medication.

International Physical Activity Questionnaire

The IPAQ-SF, which has been validated and successfully administered to similar groups (e.g. female clinical groups (Russo et al. 2018)), was identified as the most suitable, validated, clinical tool for assessing contemporaneous self-reported habitual PA and exercise at a population observation level (Craig et al. 2003). The IPAQ-SF requires participants to report how many days they walked and did moderate or vigorous PA for at least 10 min at a time in 7 days prior to completing the questionnaire. PA measured via the IPAQ-SF is classified as high, moderate, and low. High-level PA was defined as a vigorous activity, of at least 20 min per day, occurring on \geq 3 days per week, achieving a minimum of 1500 metabolic equivalent (MET) minutes per week in total or 7 days of any combination of walking, moderate-intensity or vigorous intensity activities, achieving a minimum of 3000 MET minutes per week in total. Moderate PA was defined as any of the following: vigorous activity, of at least 20 min per day, occurring on \geq 3 days per week; \geq 5 days of moderate-intensity activity and/or walking of at least 30 min per day; ≥ 5 days of any combination of walking, moderate-intensity or vigorous intensity activities; all achieving a minimum of at least 600 MET minutes per week in total. Participants were classified as low PA if they did not achieve moderate or high levels of PA (Cheng 2016).

Further research

Women were also asked if they would be interested in participating in a further research study examining the effects of PA on fertility outcomes.

Statistical analysis

Initially, summary data were extracted and reported; mean and s.D. for continuous data and percentages for categorical data were used. Correlations between PA and lifestyle factors were assessed using Pearson's correlation coefficient. The alpha level of statistical significance was set at P < 0.05. Statistical analysis was performed using SPSS for Windows (Version 26.0, SPSS). A response rate of 71.4% (250/350) was observed in this study, but not all participating women completed all sections of the demographic data collection and IPAQ-SF.

Descriptive characteristics

A total of 250 data sets were analysed from respondents (age: 34 ± 5 years, BMI: 29 ± 7 kg/m²) (Table 1). Sixty percent (*n*=143) of the women never had a pregnancy, and 82% (*n*=199) did not have children. Most women (64.4%, *n*=161) had no diagnosed cause for infertility (first tertiary clinic appointment and therefore most not had investigations). The common causes documented included: PCOS (15.2%, *n*=38), male factor (9.2%, *n*=23), endometriosis (3.6%,

Table 1	Participant demographics. Data are presented as <i>n</i>
(%) or as mean ± s.D.	

Demographic	Values
Participants, n	250
Age (years)	34 ± 5
BMI* (kg/m²)	29 ± 7
Underweight (*18.5)	1 (1%)
Normal weight (≥18.5, to ≤24.9)	81 (37%)
Overweight (≥25 to ≤29.9)	54 (24%)
Obese class 1 (≥30 to ≤34.9)	47 (21%)
Obese class 2 (≥35 to ≤39.9)	19 (9%)
Obese class 3 (≥40²)	17 (8%)
Smoker*	
Yes	24 (10%)
No	223 (90%)
Alcohol intake*	
Yes	145 (59%)
No	102 (41%)
Regular periods*	
Yes	184 (76%)
No	58 (24%)
Number of pregnancies*	1 ± 1
Number of children*	0 ± 1
Time trying to conceive* (months)	32 ± 24
Less than 1 year	30 (12%)
1–2 years	83 (33%)
>2 years	77 (31%)
Not applicable**	30 (12%)
Unknown	30 (12%)
Previous fertility treatment	
None	180 (92%)
IVF	8 (3%)
Clomid	11 (4%)
IUI	1 (1%)

*Incomplete data set; **Fertility preservation, Mayer Rokitansky Kuster Hauser (MRKH), on treatment for endometriosis, paralysis, presently single, same sex relationship, male partner known to be infertile. IVF, *in vitro* fertilisation; IUI, intra uterine insemination.

(†)

CC

This work is licensed under a Creative Commons

Attribution 4.0 International License

233

n = 9), and premature ovarian failure (POF) (2.8%, n = 7). 72% (n = 180) had no past medical history of note. Commonly reported medical co-morbidities included asthma (9.2%, n = 23), depression (5.2%, n = 13), migraines (3.2%, n = 8), hypothyroid (2.8%, n = 7), and hypertension (2%, n = 5). The majority (70.4%, n = 176) of women were not taking regular, prescribed medications. Among those who reported regular medication use, antidepressants (8%, n = 20) or inhalers (6.4%, n = 16) were most commonly listed.

Habitual PA observations

Of the 229 women who fully and accurately completed the IPAQ-SF, approximately half of the participants achieved the aforementioned criteria for moderate levels of PA, while approximately a quarter of the participants achieved low or high levels(Table 2). Women who reported taking part in structured exercise (n=71, 28%) spent 172 ± 105 min/ week doing this. On average, walking was performed on 5 ± 2 days/week for 61 ± 88 min/day. Among women who reported walking ≥10 min daily (46%, n =98), walking was performed 79±103 min/day. Only 7% (n=15) of women reported walking for 0 min/week, while time spent sitting was 5 ± 4 h/day.

Correlations

Analysis of all physical activity variables, measured using the IPAQ-SF, revealed no significant correlation with demographic variables including age, BMI, units of alcohol consumed, regular periods, or time trying to conceive (P > 0.05).

Acceptability

Of the 234 women who responded to the question querying whether they would be interested in

Table 2 Physical activity levels of the participants (n = 229) determined using IPAQ-SF. Data are presented as n (%) or as mean \pm s.p.

Physical activity outcomes	Levels
Physical activity level	
High	58 (25%)
Moderate	105 (46%)
Low	66 (29%)
Total physical activity (min/week)	87 ± 64
Total physical activity (MET-min/week)	1607 ± 1513
Vigorous activity (MET-min/week)	427 ± 669
Moderate activity (MET-min/week)	278 ± 535
Walking activity (MET-min/week)	902 ± 961

https://raf.bioscientifica.com https://doi.org/10.1530/RAF-22-0067 © 2022 The authors Published by Bioscientifica Ltd participating in future research investigating the effect of PA on fertility, 31.2% (n=73) answered yes. Among those who answered yes, women recorded the maximum amount of time they would be happy to delay fertility treatment if they took part in a future research trial (Table 3).

Discussion

Habitual PA in women experiencing infertility is unknown. Furthermore, PA is neither currently considered during routine fertility care consultations nor are there any clinical guidelines available to advise women aiming to conceive regarding safe frequency, duration, intensity, and modes of PA. The aim of the current study was, therefore, to determine habitual PA in a sample of women attending the OSI clinic. Our study found a wide variation of PA levels and types, with some women reporting performing vigorous exercise on ≥ 5 days per week (8%, n = 17), and a large proportion of women (29%, n = 66) stating that they do no moderate or high-intensity activities across the whole week, while most women reporting no structured exercise (72%, n = 179).

Evidence is available from large prospective observational studies informing that vigorous PA could be detrimental to fecundability in a population of women, with a normal BMI, not seeking fertility treatment (Wise et al. 2012, McKinnon et al. 2016), and change in menstrual cycle is also seen in women participating in vigorous exercise (Bullen et al. 1985, Ellison & Lager 1985, De Souza et al. 1998). Despite this, there is a lack of robust evidence and guidelines relating to women seeking fertility treatment to advise on how best to incorporate PA into their daily routine. Importantly, a considerable proportion of women in our cohort were outside the healthy BMI range (63%, n = 138), which has consistently been demonstrated within the literature to decrease fecundability, as well as increase the risk of miscarriage and pregnancy complications (Satpathy et al. 2008, Silvestris

Table 3 Preferred maximum time of delay to fertility treatment among women interested in participating in future research (respondents, *n* = 73).

Delay to fertility treatment	n (%)
1 month	28 (38)
3 months	8 (11)
6 months	6 (8)
12 months	12 (17)
Not sure	19 (26)



This work is licensed under a Creative Commons Attribution 4.0 International License.

3:3

Reproduction @Fertility

et al. 2018, Cozzolino *et al.* 2021). Therefore, in the context of increasing rates of women seeking fertility support, it is important to understand whether habitual PA could act as a non-pharmacological aid, with or without weight loss, and ensure that we are providing our patients with the best evidence-based care.

A recent systematic review has shown that PA may improve pregnancy rates in women with reproductive health problems, but the mode, intensity, frequency, and duration of optimal PA intervention, and the role of PA independent of weight loss, remains unclear (Mena et al. 2019). In a cohort of obese women undergoing IVF, those who performed regular PA prior to assisted reproduction experienced higher success rates, (39% vs 16% in sedentary women, P = 0.002) and improved live birth rates (24.4% vs 7.4%, P = 0.004) compared with women who did not, irrespective of weight loss (Palomba et al. 2014). This evidence provides encouraging support for the use of PA recommendations/ guidelines for obese women, who are aiming to conceive. Furthermore, a 2002 questionnaire-based prospective cohort study, originally designed to examine the association between lifestyle (smoking/exercise) and diet with breast cancer and other major diseases, found that each hour of vigorous activity undertaken per week was associated with a 7% lower relative risk of ovulatory infertility (5% on adjustment of BMI) (Rich-Edwards et al. 2002). This is in contrast to other studies examining ovulatory infertility and exercise (Bullen et al. 1985, Ellison & Lager 1985, De Souza et al. 1998) and should perhaps be interpreted with caution as the outcomes do not relate to the original question asked within the study.

PCOS is recognised as the leading cause of anovulatory infertility and affects 12-18% of women of reproductive age (March et al. 2010), with obesity prevalent in 40-88% of affected women (Balen et al. 1995, Azziz et al. 2004, Lim et al. 2012). Lifestyle modification, including PA, has been seen as a first-line approach in PCOS management (Teede et al. 2011). In a systematic review, women with PCOS were shown to have higher pregnancy rates (RR: 1.59, 95%) CI: 1.06, 2.38; five studies) and live birth rates (RR: 2.45, 95% CI: 1.24, 4.83; two observations) in PA intervention groups, compared with non-intervention controls. Further analysis showed no significant differences in improving menstrual regularity, ovulation, and conception rates between the PA intervention and comparison groups (Mena et al. 2019). Despite the high prevalence of PCOS, and the indisputable link with fertility, no robust evidence is currently available to recommend the optimum mode and volume of PA (Harrison et al. 2011, Lim et al. 2019, Dos Santos et al. 2020).

Over a quarter of the women in our cohort did low levels of PA and a quarter did high levels of PA, both could interestingly be impacting their fertility. A study looking at sedentary behaviour and idiopathic infertility showed that sedentary behaviour (OR: 3.61; 95% CI, 1.58, 8.24) was shown to be associated with infertility, but PA was not associated with fertility (Foucaut et al. 2019). Conversely, evidence is also available demonstrating that high-intensity exercise is detrimental to fertility, causing a decreased length of the luteal phase, anovulatory cycles, and eventually, amenorrhoea (Bullen et al. 1985, Ellison & Lager 1985, Green et al. 1986, De Souza et al. 1998, Hakimi & Cameron 2017, Huhmann 2020, Sophie Gibson et al. 2020). A large population-based health survey reported that increased frequency, duration, and intensity of PA were associated with increased subfertility. Women who were active on most days of the week were 3.2 times more likely to have fertility problems than inactive women, with exercising to exhaustion associated with 2.3 times the odds of fertility problems, vs low-intensity exercise (Gudmundsdottir et al. 2009). High-intensity exercise has also been associated with a 5-fold increase in IVF cycle cancellation, a 2.5-fold increase in failed implantation, a 30% lower chance of successful pregnancy, and a 50% reduction in live births compared with women who reported no regular PA in a large prospective study (Morris et al. 2006). A recent systematic review investigating the effect of exercise on ovulation showed an increased risk of anovulation in extremely heavy exercisers (>60 min/ day), but vigorous exercise (30-60 min/day) was associated with a reduced risk of anovulatory infertility (Hakimi & Cameron 2017).

Our study is novel, since we elicit the habitual PA of women who attend the OSI clinic and do not provide a prescribed intervention assuming that one intervention would suit all women. We included all women regardless of their BMI or any previously diagnosed pathology ensuring a representative sample of women who struggle with fertility and attend the OSI clinic. Unfortunately, questionnaires have been reported to overestimate PA in women and, as such, objective measurements have been recommended (Bell *et al.* 2013). However, they remain valuable to define the types and profiles of activity (Craig *et al.* 2003) and would provide preliminary information pertinent to this cohort of patients, thus were selected as the best methodological approach.

PA is an easily modifiable, affordable, and feasible lifestyle choice, which is currently ignored in the fertility setting and no guidance is available yet to inform either clinicians or their patients. This low-risk and low-cost



lifestyle choice has the potential to improve fertility without compromising the general wellbeing of the patient. We would like to add a caveat that a 'one size fits all' level of PA cannot be advocated, and a personalised approach, with a holistic take on lifestyle, not just on smoking and alcohol but also optimising the appropriate level/types of PA should be recommended to each woman seeking fertility treatment. We already appreciate that PA reduces weight, but in those who do not need to reduce weight, the role of PA in improving their fecundity requires further study. The sole focus on PA for weight loss, rather than the effects of PA on reproductive function itself, has resulted in the current lack of studies comparing types of PA, intensity, and their impact on female fertility. This precludes the efficient and effective use of PA as a therapeutic modality in the management of infertility. An adequately powered, optimally designed, clinical trial assessing the role of PA in altering fecundability is urgently needed to evaluate the public health impact of that approach and to gain insight into the biological mechanisms of PA on reproductive function. This is especially pertinent in the era of unknown and unfair funding for fertility treatment across different regions and countries on a background of intense interest among young people in PA and exercise training.

Declaration of interest

The authors declare that there is no conflict of interest that could be perceived as prejudicing the impartiality of the research reported.

Funding

Author contribution statement

Acknowledgements

References

- Ahrens KA, Vladutiu CJ, Mumford SL, Schliep KC, Perkins NJ, Wactawski-Wende J & Schisterman EF 2014 The effect of physical activity across the menstrual cycle on reproductive function. *Annals of Epidemiology* 24 127–134. (https://doi.org/10.1016/j. annepidem.2013.11.002)
- Ayton A 2019 The importance of restoring body fat mass in the treatment of anorexia nervosa: an expert commentary. *Journal of Population Therapeutics and Clinical Pharmacology* **26** e9–e13. (https://doi. org/10.15586/jptcp.v26i3.629)
- Azziz R, Woods KS, Reyna R, Key TJ, Knochenhauer ES & Yildiz BO 2004 The prevalence and features of the polycystic ovary syndrome in an unselected population. *Journal of Clinical Endocrinology and Metabolism* **89** 2745–2749. (https://doi.org/10.1210/jc.2003-032046)
- Balen AH, Conway GS, Kaltsas G, Techatrasak K, Manning PJ, West C & Jacobs HS 1995 Polycystic ovary syndrome: the spectrum of the disorder in 1741 patients. *Human Reproduction* **10** 2107–2111. (https://doi.org/10.1093/oxfordjournals.humrep.a136243)
- Bell R, Tennant PW, McParlin C, Pearce MS, Adamson AJ, Rankin J & Robson SC 2013 Measuring physical activity in pregnancy: a comparison of accelerometry and self-completion questionnaires in overweight and obese women. *European Journal of Obstetrics, Gynecology, and Reproductive Biology* **170** 90–95. (https://doi. org/10.1016/j.ejogrb.2013.05.018)
- Britton A 2016 Alcohol consumption for women trying to conceive. *BMJ* 354 i4540. (https://doi.org/10.1136/bmj.i4540)
- Bullen BA, Skrinar GS, Beitins IZ, von Mering G, Turnbull BA & McArthur JW 1985 Induction of menstrual disorders by strenuous exercise in untrained women. *New England Journal of Medicine* **312** 1349–1353. (https://doi.org/10.1056/NEJM198505233122103)
- Cheng H 2016 A simple, easy-to-use spreadsheet for automatic scoring of the International Physical Activity Questionnaire (IPAQ) Short Form. *Researchgate*. (https://doi.org/10.13140/RG.2.2.21067.80165)
- Cozzolino M, Garcia-Velasco JA, Meseguer M, Pellicer A & Bellver J 2021 Female obesity increases the risk of miscarriage of euploid embryos. *Fertility and Sterility* **115** 1495–1502. (https://doi. org/10.1016/j.fertnstert.2020.09.139)
- Craig CL, Marshall AL, Sjostrom M, Bauman AE, Booth ML, Ainsworth BE, Pratt M, Ekelund U, Yngve A, Sallis JF, et al. 2003 International physical activity questionnaire: 12-country reliability and validity. *Medicine and Science in Sports and Exercise* **35** 1381–1395. (https://doi.org/10.1249/01.MSS.0000078924.61453.FB)
- De Souza MJ, Miller BE, Loucks AB, Luciano AA, Pescatello LS, Campbell CG & Lasley BL 1998 High frequency of luteal phase deficiency and anovulation in recreational women runners: blunted elevation in follicle-stimulating hormone observed during lutealfollicular transition. *Journal of Clinical Endocrinology and Metabolism* 83 4220–4232. (https://doi.org/10.1210/jcem.83.12.5334)
- Dos Santos IK, Ashe MC, Cobucci RN, Soares GM, de Oliveira Maranhao TM & Dantas PMS 2020 The effect of exercise as an intervention for women with polycystic ovary syndrome: a systematic review and meta-analysis. *Medicine* **99** e19644. (https://doi. org/10.1097/MD.00000000019644)
- Ellison PT & Lager C 1985 Exercise-induced menstrual disorders. *New England Journal of Medicine* **313** 825–827. (https://doi.org/10.1056/ nejm198509263131314)
- Foucaut AM, Faure C, Julia C, Czernichow S, Levy R, Dupont C & ALIFERT Collaborative Group 2019 Sedentary behavior, physical inactivity and body composition in relation to idiopathic infertility among men and women. *PLoS ONE* **14** e0210770. (https://doi. org/10.1371/journal.pone.0210770)
- Green BB, Daling JR, Weiss NS, Liff JM & Koepsell T 1986 Exercise as a risk factor for infertility with ovulatory dysfunction. *American Journal of Public Health* 76 1432–1436. (https://doi.org/10.2105/ ajph.76.12.1432)



This work is licensed under a Creative Commons Attribution 4.0 International License.

3:3

This work was supported by the Wellbeing of Women (grant numbers RTF510 (N T and D K H) and RG1073 (D K H) and RG2137 (D K H)). N T is supported by an Academic Clinical Lectureship from the National Institute of Health Research. H A L is supported by the Hewitt Fertility Centre/Liverpool Women's Hospital NHS Trust.

Conceptualisation and ethical approval N T and D K H; N T, H A L, E O and E S collected the data, N T, M F, V S, and D K H analysed and interpreted the data and wrote the first draft of the manuscript. All authors have read and agreed to the final version of the manuscript.

The authors would like to acknowledge the women who attended the Hewitt Fertility Centre and kindly took time to complete the questionnaire. The authors would also like to thank Dr Steven Lane for his expert statistical assistance.

Reproduction © Fertility

3:3

- Gudmundsdottir SL, Flanders WD & Augestad LB 2009 Physical activity and fertility in women: the North-Trondelag Health Study. *Human Reproduction* 24 3196–3204. (https://doi.org/10.1093/humrep/ dep337)
- Hakimi O & Cameron LC 2017 Effect of exercise on ovulation: a systematic review. *Sports Medicine* **47** 1555–1567. (https://doi.org/10.1007/s40279-016-0669-8)
- Harrison CL, Lombard CB, Moran LJ & Teede HJ 2011 Exercise therapy in polycystic ovary syndrome: a systematic review. *Human Reproduction Update* **17** 171–183. (https://doi.org/10.1093/humupd/ dmq045)
- Huhmann K 2020 Menses requires energy: a review of how disordered eating, excessive exercise, and high stress lead to menstrual irregularities. *Clinical Therapeutics* 42 401–407. (https://doi. org/10.1016/j.clinthera.2020.01.016)
- Imterat M, Agarwal A, Esteves SC, Meyer J & Harlev A 2019 Impact of body mass index on female fertility and ART outcomes. *Panminerva Medica* 61 58–67. (https://doi.org/10.23736/S0031-0808.18.03490-0)
- Kite C, Lahart IM, Afzal I, Broom DR, Randeva H, Kyrou I & Brown JE 2019 Exercise, or exercise and diet for the management of polycystic ovary syndrome: a systematic review and meta-analysis. *Systematic Reviews* 8 51. (https://doi.org/10.1186/s13643-019-0962-3)
- Lim SS, Davies MJ, Norman RJ & Moran LJ 2012 Overweight, obesity and central obesity in women with polycystic ovary syndrome: a systematic review and meta-analysis. *Human Reproduction Update* **18** 618–637. (https://doi.org/10.1093/humupd/dms030)
- Lim SS, Hutchison SK, Van Ryswyk E, Norman RJ, Teede HJ & Moran LJ 2019 Lifestyle changes in women with polycystic ovary syndrome. *Cochrane Database of Systematic Reviews* **3** CD007506. (https://doi.org/10.1002/14651858.CD007506.pub4)
- March WA, Moore VM, Willson KJ, Phillips DI, Norman RJ & Davies MJ 2010 The prevalence of polycystic ovary syndrome in a community sample assessed under contrasting diagnostic criteria. *Human Reproduction* **25** 544–551. (https://doi.org/10.1093/humrep/ dep399)
- McKinnon CJ, Hatch EE, Rothman KJ, Mikkelsen EM, Wesselink AK, Hahn KA & Wise LA 2016 Body mass index, physical activity and fecundability in a North American preconception cohort study. *Fertility and Sterility* **106** 451–459. (https://doi.org/10.1016/j.fertnstert.2016.04.011)
- Mena GP, Mielke GI & Brown WJ 2019 The effect of physical activity on reproductive health outcomes in young women: a systematic review and meta-analysis. *Human Reproduction Update* **25** 541–563. (https:// doi.org/10.1093/humupd/dmz013)
- Mena GP, Mielke GI & Brown WJ 2020 Do physical activity, sitting time and body mass index affect fertility over a 15-year period in women? Data from a large population-based cohort study. *Human Reproduction* 35 676–683. (https://doi.org/10.1093/humrep/dez300)
- Morris SN, Missmer SA, Cramer DW, Powers RD, McShane PM & Hornstein MD 2006 Effects of lifetime exercise on the outcome of in vitro fertilization. *Obstetrics and Gynecology* **108** 938–945. (https://doi. org/10.1097/01.AOG.0000235704.45652.0b)
- NHS. Infertility. (https://www.nhs.uk/conditions/infertility/).
- NICE 2017 Fertility Problems: Assessment and Treatment. Clinical guideline (CG156), Published 20 Feb 2013; Updated 6 September 2017. (www. nice.org.uk/guidance/cg156)
- Palomba S, Falbo A, Valli B, Morini D, Villani MT, Nicoli A & La Sala GB 2014 Physical activity before IVF and ICSI cycles in infertile obese women: an observational cohort study. *Reproductive Biomedicine* Online 29 72–79. (https://doi.org/10.1016/j.rbmo.2014.03.006)
- Ricci E, Al Beitawi S, Cipriani S, Candiani M, Chiaffarino F, Vigano P, Noli S & Parazzini F 2017 Semen quality and alcohol intake: a systematic review and meta-analysis. *Reproductive Biomedicine Online* **34** 38–47. (https://doi.org/10.1016/j.rbmo.2016.09.012)

- Rich-Edwards JW, Spiegelman D, Garland M, Hertzmark E, Hunter DJ, Colditz GA, Willett WC, Wand H & Manson JE 2002 Physical activity, body mass index, and ovulatory disorder infertility. *Epidemiology* **13** 184–190. (https://doi.org/10.1097/00001648-200203000-00013)
- Russo LM, Whitcomb BW, Mumford SL, Hawkins M, Radin RG, Schliep KC, Silver RM, Perkins NJ, Kim K, Omosigho UR, *et al.* 2018 A prospective study of physical activity and fecundability in women with a history of pregnancy loss. *Human Reproduction* **33** 1291–1298. (https://doi.org/10.1093/humrep/dey086)
- Satpathy HK, Fleming A, Frey D, Barsoom M, Satpathy C & Khandalavala J 2008 Maternal obesity and pregnancy. *Postgraduate Medicine* **120** E01–E09. (https://doi.org/10.3810/pgm.2008.09.1920)
- Schmidt L 2006 Psychosocial burden of infertility and assisted reproduction. *Lancet* 367 379–380. (https://doi.org/10.1016/S0140-6736(06)68117-8)
- Sermondade N, Huberlant S, Bourhis-Lefebvre V, Arbo E, Gallot V, Colombani M & Freour T 2019 Female obesity is negatively associated with live birth rate following IVF: a systematic review and meta-analysis. *Human Reproduction Update* **25** 439–451. (https://doi.org/10.1093/humupd/dmz011)
- Silvestris E, de Pergola G, Rosania R & Loverro G 2018 Obesity as disruptor of the female fertility. *Reproductive Biology and Endocrinology* 16 22. (https://doi.org/10.1186/s12958-018-0336-z)
- Sophie Gibson ME, Fleming N, Zuijdwijk C & Dumont T 2020 Where have the periods gone? The evaluation and management of functional hypothalamic amenorrhea. *Journal of Clinical Research in Pediatric Endocrinology* **12** 18–27. (https://doi.org/10.4274/jcrpe. galenos.2019.2019.S0178)
- Sprung VS, Cuthbertson DJ, Pugh CJ, Aziz N, Kemp GJ, Daousi C, Green DJ, Cable NT & Jones H 2013 Exercise training in polycystic ovarian syndrome enhances flow-mediated dilation in the absence of changes in fatness. *Medicine and Science in Sports and Exercise* 45 2234–2242. (https://doi.org/10.1249/MSS.0b013e31829ba9a1)
- Stepto NK, Patten RK, Tassone EC, Misso ML, Brennan L, Boyle J, Boyle RA, Harrison CL, Hirschberg AL, Marsh K, et al. 2019 Exercise recommendations for women with polycystic ovary syndrome: is the evidence enough? Sports Medicine 49 1143–1157. (https://doi.org/10.1007/s40279-019-01133-6)
- Teede HJ, Misso ML, Deeks AA, Moran LJ, Stuckey BG, Wong JL, Norman RJ, Costello MF & Guideline Development Groups 2011 Assessment and management of polycystic ovary syndrome: summary of an evidence-based guideline. *Medical Journal of Australia* 195 S65–S112. (https://doi.org/10.5694/mja11.10915)
- Wise LA, Rothman KJ, Mikkelsen EM, Sorensen HT, Riis AH & Hatch EE 2012 A prospective cohort study of physical activity and time to pregnancy. *Fertility and Sterility* **97** 1136.e4–1142.e4. (https:// doi.org/10.1016/j.fertnstert.2012.02.025)
- Zegers-Hochschild F, Adamson GD, de Mouzon J, Ishihara O, Mansour R, Nygren K, Sullivan E, van der Poel S, International Committee for Monitoring Assisted Reproductive Technology & World Health Organization 2009 The International Committee for Monitoring Assisted Reproductive Technology (ICMART) and the World Health Organization (WHO) revised glossary on ART terminology, 2009. *Human Reproduction* **24** 2683–2687. (https://doi. org/10.1093/humrep/dep343)

Received 17 June 2022 Received in final form 13 August 2022 Accepted 8 September 2022 Accepted Manuscript published online 9 December 2022



This work is licensed under a Creative Commons Attribution 4.0 International License.